Author's response to reviews

Title: The Omega-3 fatty acids (Fish Oils) and Aspirin in Vascular access OUTcomes in Renal Disease (FAVoured) study: The updated final trial protocol and rationale of post-initiation trial modifications

Authors:

Andrea K Viecelli (andrea.viecelli@health.wa.gov.au)
Elaine Pascoe (e.pascoe@uq.edu.au)
Kevan R Polkinghorne (kevan.polkinghorne@monash.edu)
Carmel Hawley (carmel.hawley@health.qld.gov.au)
Peta-Anne Paul-Brent (p.kerr@uq.edu.au)
Sunil V Badve (sunil.badve@health.qld.gov.au)
Alan Cass (alan.cass@menzies.edu.au)
Stephane Heritier (Stephane.Heritier@monash.edu)
Peter G Kerr (Peter.Kerr@monash.edu)
Trevor A Mori (trevor.mori@uwa.edu.au)
Amanda Robertson (Amanda.Robertson@mh.org.au)
Hooi L Seong (laihooi@pd.jaring.my)
Ashley B Irish (ashley.irish@health.wa.gov.au)

Version: 2
Date: 7 June 2015

Author's response to reviews:

See attached letter